Author’s response to reviews

Title: The health disparities cancer collaborative: a case study of practice registry measurement in a quality improvement collaborative

Authors:

David A Haggstrom (dahaggst@iupui.edu)
Steven B Clauser (clauers@mail.nih.gov)
Stephen H Taplin (taplins@mail.nih.gov)

Version: 6 Date: 21 May 2010

Author’s response to reviews:

I have ensured that the Abstract present in the manuscript matches exactly what is in the submission system so that the Results section and Conclusions are identical.

MINOR:

The country has been included in the affiliation details.

Figure legends: I have tried to clarify which title/legend corresponds to which figure by consecutive numbering, but I'm uncertain if I've interpreted this instruction appropriately.